Author’s response to reviews

Title: Developing an Intervention for Fall-Related Injuries in Dementia (DIFRID): an integrated, mixed-methods approach

Authors:
Alison Wheatley (alison.wheatley@ncl.ac.uk; alison.wheatley@newcastle.ac.uk)
Claire Bamford (claire.bamford@ncl.ac.uk)
Caroline Shaw (c.a.shaw@hotmail.co.uk)
Elizabeth Flynn (Elizabeth.Flynn@nuth.nhs.uk)
Amy Smith (amy.smith16@nhs.net)
Fiona Beyer (fiona.beyer@ncl.ac.uk)
Chris Fox (Chris.fox@uea.ac.uk)
Robert Barber (Robert.Barber@ntw.nhs.uk)
Steve Parry (Steve.Parry@nuth.nhs.uk)
Denise Howel (denise.howel@ncl.ac.uk)
Tara Homer (tara.homer@ncl.ac.uk)
Louise Robinson (a.l.robinson@ncl.ac.uk)
Louise Allan (L.Allan@exeter.ac.uk)

Version: 3 Date: 11 Feb 2019

Author’s response to reviews:

1. We noticed that you stated informed consent was obtained both written and verbally. Please briefly clarify the instances when verbal consent was obtained and indicate whether an ethics committee approved this form of consent.

Response: We have added that verbal consent was obtained for the professional interviews and consensus panel. The University ethics committee gave approval for the verbal consent for the professionals. The consensus panel did not require ethical approval but consent was obtained.
2. Please also include the full name of the ethics committee at Newcastle University that approved the initial interviews and focus groups.

Response: The full name is given as Newcastle University research ethics committee.

3. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

Response: This has been provided.

4. Please include a list of the Figure legends after the References, before the Additional files list.

Response: This has been provided.

5. We noticed that several authors are missing from the Authors’ contributions section. Please ensure that all authors and their individual contributions are included in this section.

Response: All authors are now included.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: These have been provided.